**Author's response to reviews**

**Title:** "The sun keeps rising but darkness surrounds us": a qualitative exploration of the lived experiences of women with obstetric fistula in Ethiopia

**Authors:**

shifera yedenekal (shiferaa@yahoo.com)

Misganaw Animut (misganawanimut@gmail.com)

Abebe Gebretsadik (abema24@yahoo.com)

Lakew Gebretsadik (lakewgebretsadika9@gmail.com)

Million Berhe (megaabera@yahoo.com)

Zewdie Koricha (zbkoricha@yahoo.com)

**Version:** 9  **Date:** 04 Feb 2019

**Author's response to reviews:**

Response for editor.

1. Ethics Approval and Consent to Publish?

Response: Written consent was taken.